British Society for Surgery of the Hand Evidence for Surgical Treatment (BEST)

Topic: Evidence based management of adults with acute ulnar collateral ligament (UCL) of the thumb injuries

Date of publication: April 2023

Date of anticipated review: 2028



NICE has accredited the process used by **British Society for Surgery of the Hand** to produce Clinical Guidelines. Accreditation is valid for 5 years from XXXX. More information on NICE accreditation can be viewed at <u>www.nice.org.uk/accreditation</u>.

Contents

AUTHORS
GUIDELINE DEVELOPMENT GROUP (GDG):6
ROLES OF AUTHORS WHO ARE NOT GDG MEMBERS6
ACKNOWLEDGEMENTS6
FUNDING SOURCES USED6
COLLABORATING ORGANISATIONS7
CONFLICTS OF INTEREST
DISCLAIMER8
PROCESS
OVERALL OBJECTIVE9
ANTICIPATED USERS9
TARGET POPULATION9
QUESTIONS DISCUSSED IN THIS BEST9
INCLUSION & EXCLUSION CRITERIA10

PLAIN LANGUAGE SUMMARY	10
INTRODUCTION	11
METHODS	12
SYSTEMATIC REVIEW RESULTS	14
OTHER STUDIES (NOT INCLUDED IN THE SYSTEMATIC REVIEW):	18
SYSTEMATIC REVIEW OVERVIEW DISCUSSION:	19
CLINICAL PRACTICE RECOMMENDATIONS:	20
GOOD PRACTICE POINTS:	20
CLINICAL AUDIT INDICATORS:	21
RESOURCE IMPLICATIONS:	21
FACILITATORS AND BARRIERS TO IMPLEMENTATION:	21
FUTURE RESEARCH RECOMMENDATIONS:	21
STAKEHOLDERS INVITED TO PROVIDE EXTERNAL REVIEW:	23
TIMELINE OF GUIDELINE:	23
APPENDIX 1: PRISMA FLOW CHARTS FOR SYSTEMATIC REVIEW	24

APPENDIX 2: REVIEW TABLES INCLUDING STUDY CHARACTERISTICS AND EVIDENCE	26
APPENDIX 3: QUALITY ASSESSMENTS	41
APPENDIX 4: KEY CLINICAL PRACTICE RECOMMENDATIONS	50
APPENDIX 5: PATIENT FLOW ALGORITHM	51
APPENDIX 6: SUPPORT TOOL: QUICK REFERENCE GUIDE	53
APPENDIX 7: REFERENCES	55

Authors

Benjamin Dean BM BCh DPhil FRCS (Trauma & Orth). Consultant Hand Surgeon and Senior Research Fellow, Oxford University Hospitals NHS Trust, Benjamin.dean@ndorms.ox.ac.uk

Jeremy Rodrigues FRCS (Plast). Consultant Plastic Surgeon and Senior Lecturer, University of Warwick, Jeremy.rodrigues@warwick.ac.uk

Nicholas Riley FRCS (Trauma & Orth). Consultant Hand Surgeon, Oxford University Hospitals NHS Trust, Nicholas.riley@ouh.nhs.uk

Nicholas Rabey FRCS (Plast), Consultant Plastic Surgeon, University Hospitals of Coventry and Warwickshire, nick.rabey@gmail.com

Ella Donnison. Hand therapist, Pulvertaft Hand Centre, Derby, ella.donnison@nhs.net

Kirsty Challen, Consultant in Emergency Medicine at Lancashire Teaching Hospitals NHS Foundation Trust, Kirsty.Challen@lthtr.nhs.uk

Steph Bradford, Patient representative

Guideline Development Group (GDG):

Core Stakeholders:	Benjamin Dean, GDG Lead
	Jeremy Rodrigues, Representing Hand Surgeons
	Rowa Taha, Representing Hand Surgery Trainees
	Nicholas Rabey, Representing Hand Surgery Trainees
	Nicholas Riley, Representing Hand Surgeons
	Kirstye Challen, Representing Emergency Medicine
	Kannan Rajesparan, Representing Radiology
	Ella Donnison, Representing Hand Therapists
	Katie Sehmi, Representing Hand Therapists
	Steph Bradford, Representing Patients
	Donna Harrison, Representing Patients
Special Stakeholders:	Taz Syed, Representing Commissioners

Roles of authors who are not GDG members

Sally Hopewell: Systematic review expertise Neal Thurley: Research librarian and carried out literature searches

Acknowledgements

Funding sources used

None

Collaborating organisations

This guideline has been reviewed and endorsed by the Councils of the British Society of Skeletal Radiologists (BSSR), the British Association of Hand Therapists (BAHT) and the Royal College of Emergency Medicine (RCEM).

Conflicts of interest

There are no conflicts of interest of relevance for any of the GDG members or document authors.

Disclaimer

This document reflects a consensus view of the British Society for Surgery of the Hand Research Committee and Council, based on a systematic and transparent review of evidence. All users of this document must ensure that they consider the entirety of the document when using it, that the recommendations within this guideline are not mandatory, and that clinical judgement and that patient-centred decision making for all individual patients is the highest priority. Users are reminded of their individual duties and responsibilities, professional or otherwise, to use this guideline responsibly and that no content within this guideline overrides these duties and responsibilities.

Process

This document has been produced by systematic reviews, with the interpretation and development of recommendations achieved by consensus of the GDG members.

Overall Objective

The overall objective is to describe the management of adult patients with acute ulnar collateral ligament (UCL) injuries of the thumb metacarpophalangeal joint (MCPJ) in the United Kingdom.

Anticipated Users

The anticipated users are health care professionals treating patients with acute UCL injuries, those commissioning care for patients, and possibly patients and carers of patients with acute UCL injuries.

Target Population

Adults with acute UCL injuries of the thumb MCPJ.

Questions discussed in this BEST

- 1. How should patients with suspected UCL injuries be initially assessed, investigated and managed?
- 2. Which patients should be referred to specialist services and when should they be seen?
- 3. How should patients be further assessed and investigated by specialist services?
- 4. What treatments should be offered to which patients and when?
- 5. Which treatments are superior to other treatments?
- 6. Which treatments are more cost-effective than other treatments?
- 7. What outcomes can be expected from specific treatments?
- 8. What future quality improvement work and/or research might be beneficial in this area?

Questions not discussed in this BEST

1. Questions relating to non-acute (chronic) UCL injuries

Inclusion & exclusion criteria

Patients 18 years and older (adults) with acute injuries (3 or less weeks from injury) to the ulnar collateral of the thumb MCPJ were included. Paediatric UCL injuries were not considered as part of this systematic review.

Plain Language Summary

Sprains and tears of the thumb ligaments are a relatively common injury. The ulnar collateral ligament (UCL) is the ligament closest to the webspace of the thumb's metacarpophalangeal joint (MCPJ- this is the joint which makes up the largest 'knuckle' of the thumb). The UCL is frequently injured, resulting in pain and difficulty using the thumb, particularly when pinching. Patients with these injuries mostly go to Emergency Departments (EDs) and Minor Injuries Units (MIUs), and they are then often referred onto hospital services (secondary care) for further treatment. Patients may be treated with non-surgical treatments such as splints and plaster casts to keep the thumb MCPJ still, or with surgery to repair or reconstruct the ligament. We searched for all studies that assessed the diagnosis of UCL injuries and that assessed the treatment of UCL injuries. A group that included two patients, a radiologist, a commissioner, an emergency medicine doctor, hand therapists and surgeons then formally discussed the studies in order to agree upon recommendations of how to diagnose and treat UCL injuries.

The group's recommendations are that patients with acute UCL injuries should be assessed with a history, clinical examination, and x-rays. Patients without significant joint laxity can be treated non-surgically. Patients with significant joint laxity on clinical examination may be treated with non-surgical joint immobilisation or surgical repair and should reach a shared decision with their clinician about the definitive treatment within 2 weeks of presentation.

Introduction

Acute complete ruptures of the ulnar collateral ligament (UCL) of the thumb metacarpophalangeal joint (MCPJ) are common injuries, accounting for around 50 in 100,000 presentations to Emergency Departments (EDs)(1). These injuries frequently result in pain and dysfunction, which can be persistent in a minority of cases. More minor 'sprains' without joint instability on clinical examination are generally treated with early movement as pain allows, while there is more controversy as to how best to manage complete UCL ruptures which typically manifest with joint instability on clinical examination.

Understanding the anatomy of the region and the variation in severity of the injury is of clinical importance. Clinical examination requires an appreciation of the pathophysiology of the adductor muscle aponeurosis, proper UCL, accessory UCL and palmar plate. The MCPJ is typically examined by applying a valgus force in extension and in a degree of MCPJ flexion to relax the palmar plate, the latter thought to be isolating the UCL proper. The clinical investigations for UCL injuries include simple xrays, x-rays while applying a force to the MCPJ (stress x-rays), Ultrasound (USS) and Magnetic Resonance Imaging (MRI). The term 'Stener lesion' is used to describe when the ligament is completely torn and is retracted, allowing the adductor aponeurosis to become interposed between the torn ligament and its site of bony insertion. There is some controversy as regards the 'Stener lesion' relating to its true frequency, how best to diagnose its presence and how best to treat it.

Our aim was to perform two systematic reviews. The first 'diagnostic' review aimed to assess the strength of evidence relating to the diagnosis of acute UCL injuries. The second 'therapeutic' review aimed to assess the strength of evidence relating to non-surgical and surgical interventions for acute UCL injuries.

11

Methods

Both systematic reviews are carried out according to PRISMA guidance with a preregistered protocol, searches carried out by the research librarian, and two reviewers performing the screening and data output.

Diagnostic:

The inclusion was any study relating to acute UCL injuries in adults (aged \geq 18). Exclusion criteria included age <18 years, chronic injuries (presentation>3weeks from injury), injuries with a significant bony avulsion fragment, open injuries, and nonisolated UCL injuries. Review articles, studies not published as a full article (conference abstracts), studies not involving patients and case studies were also excluded. The intervention was patients undergoing any form of diagnostic test relating to acute suspected UCL injuries. This included any retrospective and prospective cohort/case control studies, and randomised controlled trials (all types of randomised trials were included). The initial search yielded 1579 articles. After screening by title, abstract and removing duplicate and non-English studies, 88 studies were selected for further full text assessment of eligibility by the authors. Following this, 24 studies were finally selected as being relevant to the research question (Appendix 1 – Figure 1). Six studies assessed clinical examination, 12 US, six MRI and two stress arthrography. One study assessed both ultrasound and MRI as index tests and another both clinical stress testing and ultrasound. The reference tests used were surgery, MRI and clinical follow-up. The study quality was assessed using the QUADAS-2 tool(2). Levels of evidence were grade from 1 to 4 based on The Oxford Centre for Evidence-based Medicine - Levels of Evidence.

Therapeutic:

The inclusion was any study relating to acute UCL injuries in adults (aged \geq 18). Exclusion criteria included age <18 years, chronic injuries (presentation>3weeks from injury), injuries with a significant bony avulsion fragment, open injuries, and non-isolated UCL injuries. Review articles, studies not published as a full article (conference abstracts), studies not involving patients and case studies were also excluded. The intervention was patients undergoing any form of therapeutic (surgical or non-surgical) intervention and the comparator was any therapeutic intervention

12

(including, but not be restricted to, e.g. active monitoring, usual care, non-surgical interventions such as early mobilisation or splinting or cast treatment, surgical interventions or similar). The study design had to include an intervention and a comparator. This included any retrospective and prospective cohort/case control studies, and randomised controlled trials (all types of randomised trials were included). A total of 1161 records were identified through database searching (Appendix 1 – Figure 2). After removal of duplicate entries, 761 titles and abstracts were screened, 11 full articles were assessed and six met the inclusion criteria. Shortlisted studies were assessed using SIGN50 methodology.

Systematic review results

Diagnostic

Clinical examination:

Of the six studies which reviewed clinical examination techniques, two were of level 2 evidence and four of level 3 (Appendix 2 -Table 1)(3–8). The level 2 studies tested cohorts of 23 and 30 patients respectively and both stated a positive clinical diagnosis of a valgus deformity of > 35° on stress testing However, Heyman et al. tested patients under local anaesthetic block with the MCPJ in 30° flexion and palpated for a mass, and Mahajan et al. clinically assessed with the MCPJ in 30° flexion and extension whilst noting a fixed end-point with no mention of local anaesthesia(6,8). Heyman et al. used surgery as the reference standard and documented displaced ligaments alone, with the results giving a sensitivity of 0.94 (95% CI 0.70 to 1.00) and specificity of 0.57 (95% CI 0.18 to 0.90). In comparison, Mahajan et al. used MRI as the reference standard and assessed both displaced and ruptured undisplaced ligaments. Their results gave a sensitivity of 0.92 (95% CI 0.64 to 1.00) and specificity of 0.91 (95% CI 0.71 to 0.99) and specificity of and 0.75 (95% CI 0.35 to 0.97) for ruptured undisplaced ligaments.

The remaining four level 3 studies used differing criteria for a positive diagnosis of injury. These included the presence of a palpable 'tender tumor', valgus stress instability testing with no specified threshold angle or degree of thumb flexion, a laxity of >15° compared to the contralateral thumb and Cooper et al. compared examination either with or without local anaesthesia (3–5,9). Louis et al. compared two protocols of examination from having no specified angle of laxity for the first 20 patients to >35° with the thumb in full flexion for the later 20 patients. The reference standards also varied with one study using stress radiography alone, one using surgery only and two using a combination of surgery and clinical follow-up. Two studies were considered to have an unclear risk of bias and the remaining four a high risk of bias for the reference standard (Appendix 3). Further statistical analysis was not carried out due to the heterogeneity of studies in regards the differing examination techniques and inconsistent reference tests.

Ultrasound:

The search identified 12 studies which assessed the use of ultrasound in the diagnosis of UCL injuries. Of these, two were level 1 studies, four level 2 studies and six level 3 studies (Appendix 2 -Table 2). The publication dates of the studies ranged from 1993 to 2018. Of the two level 1 studies, Shekarchi et al. compared the accuracy of diagnosing complete UCL rupture in 20 patients on ultrasound with a 12MHz probe with the findings confirmed on MRI, giving a sensitivity of 0.71 (95% CI 0.29 to 0.96) and specificity of 0.85 (95% CI 0.55 to 0.98)(10). In contrast, Susic et al. did not report the specifications of the ultrasound used to assess a completely torn and displaced UCL in 14 patients with clinical signs of injury and confirmed findings at surgical exploration(11). This study gave figures reporting a sensitivity of 0.40 (95% CI 0.05 to 0.85), specificity of 0.78 (95% CI 0.40 to 0.97) and accuracy of 0.64 (95% CI 0.35 to 0.87) for a completely torn and displaced ligament.

In the four level 2 studies, the ultrasound transducer frequency was not reported in one, 7.5 MHz in two and a range of 10 to 17 MHz in the other(12–14). All four studies compared findings using surgery as the reference standard although one of the Hergan et al. studies also reviewed the use of MRI. Chuter et al. tested the diagnosis of all complete ruptures, whereas the two studies by Hergan et al. and Melville et al. assessed the diagnosis of displaced ligaments. The six remaining trials of level 3 evidence used as their reference standards both surgery and clinical follow-up, meaning no comparable values for sensitivity or specificity could effectively be described (5,15–19). Comparing the results of the four studies which assessed the effectiveness of ultrasound in diagnosing completely torn and displaced ligaments using surgery as the reference standard, gives a range of test sensitivity from 0.4 to 1.00, specificity from 0.78 to 1.00 and accuracy from 0.64 to 1.00 from a total of 96 participants.

The studies were reviewed for their individual risk of bias and applicability. All were scored as having low concerns of applicability for the patient selection, index test, and all but one for reference test. However, 33% were considered to have a high risk of bias and 50% an unclear risk of bias for the reference standard. When flow and timing was considered, 42% of studies had a high risk of bias and 50% an unclear risk of bias (Appendix 3).

MRI:

Six trials were identified pertaining to the use of MRI (Appendix 2 -Table 3). Of these, one had an evidence level of 2 and the rest were level 3. The level 2 trial additionally considered the use of ultrasound and adopted surgery as the reference standard (13). All five of the level 3 trials compared both surgery and clinical follow-up as the reference standards. Only the level 2 trial provided sufficient data for a statistical test result, reporting a sensitivity of 1.00 (95% CI 0.54 to 1.00) and specificity of 1.00 (95% CI 0.72 to 1.00) for displaced and non-displaced tears in a sample of 17 patients.

In the five level 3 trials a total of 83 patients underwent MRI and 14 MR arthrography (20–24). Of these, 49 (51%) were confirmed with surgery and the rest were treated conservatively with clinical follow-up. Out of all the studies there was one reported instance of confirmed incorrect diagnosis with MRI, whereby a UCL diagnosed as moderately displaced was found to have a partial tear at surgery(24).

When the risk of bias and applicability concerns are considered, none of the studies were found to have a low risk of bias for the reference test or flow and timing domains. This was either because the patient sample outcome was assessed with two distinct reference tests or the details of how the reference test was carried out and whether the assessors were blinded was not reported. Adequate detail was given for the conduct of MRI in 50% of the studies. However, there were problematic or unclear patient selection methods in five of the six studies (Appendix 3).

Therapeutic

The results are summarised in Tables 4, 5 and 6. The study by Sollerman et al compared a functional splint with plaster cast treatment in patients with complete UCL ruptures(25); patients were managed both surgically and non-surgically. The authors reported no difference in MCPJ range of movement (ROM), grip strength and sick leave taken; however, the data provided were insufficient for any further analysis, such as a forest plot. The RCT by Rocchi et al. compared the outcomes of operated patients treated with either a traditional standard thumb spica which immobilized the MCPJ or a new modified thumb spica which allowed early MCP motion(26). At 12 months the new spica group had increased MCPJ ROM (standardized mean difference (SMD), -3.69; 95% confidence interval (CI), -2.46--4.92, P<0.0001), a better Dreiser index (SMD, 1.65; 95%CI, 0.81–2.50; P=0.0001) and reduced pain VAS (SMD, 1.53; 95% CI, 0.70–2.35; P=0.0003). There was no statistically significant difference between groups in tip pinch strength at any time point. The RCT by Crowley et al. compared outcomes between patients treated with early active mobilization or plaster immobilization after being treated surgically with Mitek anchor repair(27). The outcome data was not provided, meaning that any further analysis was not possible. The retrospective comparative case series by Saetta et al. demonstrated a higher chance of an excellent functional result with suture repair versus steel wire, but this was not statistically significant (risk ratio, 1.19; 95% Cl, 0.82–1.71); the other outcome data was incomplete and thus precluded further analysis(28). The retrospective case series by Lane demonstrated no statistically significant difference in the chances of a full versus partial recovery in ROM of the MCPJ, of a full versus partial recovery in strength and of a full versus partial functional recovery(29). The study by Katolik et al. did not provide adequate data with which to conduct any further analysis(30).

Rocchi et al. demonstrated no statistically significant difference in complication rate between treatment with the standard spica and the new spica (risk ratio, 1.5; 95% Cl, 0.29–7.73); the complications consisted of three cases of temporary dysaesthesia and two cases of inflammatory scars. The complication rate was identical in both the early active mobilization and plaster cast groups in the study by Crowley et al. (Risk ratio: 1.0, 95% CI: 0.32, 3.10); all six complications in this study were that of scar tethering, with all resolving with ultrasound therapy and massage. The studies by Saetta et al. and Sollerman et al. did not make any mention of specific complications. Lane et al demonstrated no statistically significant difference in the complication rate between the older method of pull out suture plus K-wire fixation and the new method of suture repair (risk ratio, 3.57; 95% CI, 0.25–50.15); there was one complication with the traditional method (broken pull-out suture at 2 weeks) and one with the new method (re-rupture at 9 months) The study by Katolik et al demonstrated a higher complication rate with pull-out suture versus bone anchor repair. but this was not statistically significant (risk ratio, 4.00; 95% CI, 0.92–17.30); all the ten complications were soft-tissue-related (five were persistent wound erythema consistent with wound infection and five were paraesthesiae, which resolved over time).

Overall, all studies were deemed to be at a high risk of bias, particularly in terms of blinding of outcome assessment and selecting reporting. There is a lack of high quality prospective studies using reliable and valid patient reported outcome measures (PROMs). Only the study by Rocchi used a validated PROM, and none of the other studies used validated PROMS

Other studies (not included in the systematic review):

There are no randomised controlled trials in the English literature on other aspects of diagnosis or management which fit the inclusion criteria of this systematic review. However, there are some cohort studies which are worth mentioning:

Pichora et al(31)

This cohort study followed up 32 patients who had been investigated with stress radiography, arthrography and clinical examination. All patients were treated with a removable custom splint. In the 32 patients available for follow-up, mean relative instability improved from 17 degrees after injury to 2.3 degrees at follow up. Functional and subjective outcomes were good or satisfactory in more than 90% of patients, pinch strength recovered to 89% of the contralateral thumb at approximately 1 year following injury. Outcomes for all patients with Stener lesions were satisfactory, although joint stability was less than in the whole group. The three failures involved persistent symptoms, that defied subsequent surgery and which were not related to joint instability.

Landsman et al(32)

This cohort study assessed the outcomes in 39 patients with 40 UCL ruptures. All patients were assessed clinically and deemed to have significant laxity of the UCL. All patients were treated with splint immobilisation for a minimum of 8 weeks and six patients (15%) were treated with delayed surgery as a result of no firm endpoint on stressing the UCL. In the 34 patients who did not require delayed surgery there was a recovery of pinch strength to 92% of the contralateral thumb at beyond 1 year. Also 28 patients of 34 had no pain on daily activities, while the remaining 6 patients reported an occasional ache on strenuous activities.

Systematic review overview discussion:

Diagnostic

The six clinical assessment studies reported sensitivities between 0.91 and 0.94 and specificities between 0.41 and 0.75. Twelve ultrasound studies stated sensitivities between 0.4 and 1.0 and specificities between 0.78 and 1.0. From six MRI studies, one stated a sensitivity of 1.0 and specificity of 1.0. However, when the studies were assessed using the QUADAS-2 tool, most were determined to be of low to moderate quality with significant heterogeneity in design. Despite the term 'Stener lesion' being widely used, no study has demonstrated that it can be reliably diagnosed by any form of clinical examination or investigation. Overall, these results support the use of clinical examination given its high sensitivity for the detection of displaced ligaments; however the role of ultrasound and MRI remains unclear. In summary significant laxity on clinical examination is defined as any of the following: 1) No firm end point to stressing of the UCL in full extension or 30 degrees of flexion of the MCPJ; 2) >20 degrees more laxity than the contralateral thumb in full extension or 30 degrees of flexion of the MCPJ or 30 degrees of flexion of MCPJ.

Therapeutic

There is some low quality evidence which supports early mobilisation for surgically treated patients. The studies by Crowley et al and Rocchi et al have demonstrated some early functional benefits to early mobilisation after surgery. Therefore, there is sufficient evidence to support early mobilisation after surgery when it is felt to be safe to do so.

The natural history of complete ruptures is uncertain as the rate of failure of nonsurgical treatment is highly variable. Pichora et al reported only 3 failures (7%) of nonsurgical treatment in 42 patients, and all these failures were not associated with joint instability(31). Landsman et al reported 6 failures (15%) out of 40 patients which were all associated with joint instability(32). While Milner et al reported a very low rate of failure for ruptures with less than 3mm displacement on MRI and a 90% failure rate for ruptures with >3mm displacement(23).

There is a lack of evidence comparing surgery to non-surgical treatment, and no prospective study has compared surgery to non-surgical treatment. Non-surgical immobilisation can be either with a cast or a customised splint, however there is no evidence to demonstrate the superiority of a specific type of immobilisation.

There is also a high level of uncertainty relating to the outcomes for both non-surgical and surgical treatments due to the lack of high quality evidence. There is a lack of high quality prospective studies using reliable and valid patient reported outcome measures (PROMS). Of note only one of all the included therapeutic studies used a validated PROM.

Clinical practice recommendations:

Based on the current available evidence, clinical examination is recommended to assess for significant laxity of the UCL (**Iow evidence**). There is insufficient evidence to mandate the routine use of ultrasound (USS) or magnetic resonance imaging (MRI).

Patients without significant joint laxity should be treated non-surgically. It is reasonable to offer early surgery or non-surgical immobilisation of the MCPJ to patients with significant joint laxity on clinical examination (**very low evidence**).

Good practice points:

It is considered good practice that:

- Patients are assessed by history, clinical examination and x-rays in two orthogonal planes. This initial clinical examination should be performed by an appropriately trained healthcare professional
- There is a local pathway for the management of suspected UCL injuries which involves specialist musculoskeletal (MSK) services and access to definitive surgical care when deemed necessary
- Patients with pain but preserved function AND no clinical evidence of significant joint laxity AND normal x-rays may be discharged with safety net advice
- Patients who do not meet the above criteria for early discharge should be referred on to specialist MSK services
- A shared decision about definitive management should be reached within 2 weeks of a patient's referral to specialist MSK services (the specialist MSK services should be capable of providing surgery when needed)
- Non-surgical immobilisation for patients with significant joint laxity should be with a rigid orthosis such as a cast or thermoplastic splint

Clinical audit indicators:

It is considered that the following could be used as clinical audit indicators:

- Suitable validated patient reported outcome measures (PROMs)
- Pinch strength
- Persistent joint instability

Resource Implications:

It is believed that the clinical practice recommendations and good practice points align with existing NHS practice. Therefore, the resource implication of implementing this guideline is considered minimal.

Facilitators and barriers to implementation:

If clinical staff are not competent in assessing UCL injuries, then training may be required. Such training is not believed to be complex, expensive or onerous to deliver. No other significant barriers to implementation have been identified. It is suggested that using the quick reference as a standalone reference may be facilitator. For example, users may wish to make the quick reference guide could be made available in clinical areas.

Future research recommendations:

Areas for future research into the management of UCL injuries include:

- High quality prospective cohort studies to better understand the natural history of UCL injuries
- High quality diagnostic studies to assess the reliability and validity of modern imaging techniques, as well as how these relate to clinical prognosis

- High quality RCTs to investigate the clinical and cost effectiveness of surgery versus non-surgical joint immobilization
- High quality RCTs to assess the clinical and cost effectiveness of different rehabilitation regimes after surgery

(It should be noted that PROMs should be an integral part of any future research studies and that a diagnostic study could potentially be embedded within a future RCT)

Stakeholders invited to provide external review:

The British Orthopaedic Association The British Association of Plastic, Reconstructive and Aesthetic Surgeons The British Association of Hand Therapists The Royal College of Emergency Medicine The Royal College of Radiologists

Timeline of guideline:

Date topic identified: Date GDG lead appointed: Date draft supplied by GDG authors: Date Internal review completed: Dates of public consultation: Date external review completed: Date published:

Appendix 1: PRISMA flow charts for systematic review



Figure 1: Diagnostic review flow chart





Appendix 2: Review tables including study characteristics and evidence

Table 1: Characteristics of included studies- Clinical Examination

Study (Year)	Country	Level of	Number	Mean	Index	Reference	Outcome summary	Sensitivity	Specificity
	of Origin	Evidence	of	age	Test(s)	Test(s)			
			patients	(range)					
Abrahamsson	Sweden	3	24	27 (15-49)	'Palpable' lump	Surgery (8) /	7 of 8 patients with a lump had displaced UCL	n/a	n/a
et al. (1990)					at MCPJ	1-year clinical	at surgery. 15 of 16 patients with no lump		
						stability (16)	treated conservatively had a stable MCPJ.		
Cooper et al.	UK	3	47	32.4 (14-73)	Clinical stress	Stress	39 patients had negative stress tests for both	87.5%	100%
(2005)					test under local	radiography	index and reference assessment. 7 had	(laxity)	(laxity)
					anaesthetic		positive assessments for both. 1 patient had a		
					(LA)		negative test under LA but positive stress		
							radiogram.		
Heyman et al.	US	2	23	Not reported	Clinical stress	Surgery	Valgus >35° in 30° flexion and full extension	94%	57%
(1993)					test		had a completely torn and displaced ligament	(dislocation)	(dislocation)
							in 15 of 17 patients.		

Louis et al.	US	3	2 types of	22.5 (14-29)	Clinical stress	Surgery	Technique 1: of positive patients 20% of	n/a	n/a
(1986)			examination		tests at-		ligaments were displaced at surgery.		
			techniques:		1. Extension		Technique 2: of positive patients 70% of		
			20+20		and varying		ligaments were displaced at surgery.		
					degrees of				
					flexion				
					2. >35° laxity				
					with MCPJ in full				
					flexion				
Mahajan et al.	Netherlands	2	30	Not reported	Clinical stress	MRI	MCPJ valgus >35° in extension and >20° in	91%	75%
(2016)					test		30° flexion and/or no fixed endpoint.	(complete	(complete
							- 15 patients had positive stress tests. 13 had	rupture)	rupture)
							complete UCL rupture and 9 completely torn	92%	41%
							and displaced on MRI.	(dislocation)	(dislocation)
							- 7 patients had no end-point. All had complete		
							rupture with 3 completely torn/displaced on		
							MRI.		
							- 8 patients had 'inconclusive results'. 2 were		
							ruptured, 1 completely torn and displaced, 3		
							partial tear, 3 intact.		

М	lurphey et al.	US	3	25	Not reported	Clinical Stress	Surgery (14) /	MCPJ valgus >30° +/- palpable 'Stener' lesion:	n/a	n/a
(1	1997)					Test / Stress	Clinical	- Identified 'correct grade' in 24 of 25 (96%)		
						Ultrasound	Follow-Up	patients and UCL dislocations in 5 of 8 (62%)		
							(11)	patients.		

Study	Country	Level of	Number	Mean	Index	Reference	Outcome summary	Sensitivity	Specificity
(Year)	of Origin	Evidence	of	age	Test(s)	Test(s)			
			patients	(range)					
Chuter et	UK	2	79	40 (12-81)	Ultrasound	Surgery	127 patients had surgery, of which 79 had prior	92%	n/a
al. (2009)							ultrasound. Of these, 1 was false positive diagnosis	(complete	
							and 6 were false negative. PPV was 99%.	rupture)	
Hergen et	Austria	2	39	37.2 (16-61)	Ultrasound	Surgery	36 of 39 patients had a correct preoperative diagnosis.	89%	95%
al. (1995a)							5 had no rupture, 15 complete rupture and 16	(dislocation)	(dislocation)
							dislocation. Of the 3 misdiagnosed patients one was	88%	94%
							delayed, one had a technical error and the other had a	(undisplaced	(undisplaced
							misinterpretation of the image.	rupture)	rupture)
Hergen et	Austria	2	17	37 (14-70)	Ultrasound /	Surgery	Ultrasound correctly diagnosed 15 of 17 patients. Of	83%	91%
al. (1995b)					MRI		these, 5 of 6 displaced tears and 10 of 11 undisplaced	(dislocation)	(dislocation)
							tears were confirmed at surgery.	91%	83%
								(undisplaced	(undisplaced
								rupture)	rupture)

Table 2: Characteristics of included studies- Ultrasound Assessment

Hoglund	Sweden	3	64	35 (10-81)	Ultrasound	Surgery (39) /	Ultrasound correctly diagnosed 32 of 39 patients who	n/a	n/a
et al.						Clinical Follow-	received surgery. Of these 13 of 17 undisplaced tears		
(1995)						Up (25)	and 13 of 16 completely torn and displaced tears were		
							confirmed at surgery.		
Jones et	UK	3	60	34 (11-79)	Ultrasound	Surgery (19) /	For complete UCL rupture, 17 were diagnosed	n/a	n/a
al. (2000)						Clinical Follow-	clinically and 11 of these were confirmed with both US		
						Up (41)	and surgery. 4 of these had a completely torn and		
							displaced ligament confirmed with US and surgery.		
							Of the other 6 diagnosed as intact on US, 3 had non-		
							displaced ruptures at surgery.		
Kohut et	Switzerland	3	21	27 (16-63)	Ultrasound	Surgery (7) /	7 of 21 patients had surgery. Of these, the ultrasound	n/a	n/a
al. (1993)						Clinical Follow-	findings of a completely torn and displaced ligament		
						Up (14)	matched the surgical exploration in 4 patients. 2		
							patients had a false positive result, and 0 patients false		
							negative.		
Melville et	US	2	26	40 (19-75)	Ultrasound	Surgery	26 patients had 17 completely torn and displaced	100%	100%
al. (2013)							ligaments, 7 undisplaced ruptures and 2 partial-	(dislocation)	(dislocation)
							thickness tears at surgery.		
Murphey	US	3	25	Not reported	Clinical	Surgery (14) /	Ultrasound:	n/a	n/a
et al.					Stress Test /	Clinical Follow-	- Identified 'correct grade' in 25 of 25 patients (100%)		
(1997)					Stress	Up (11)	and UCL dislocations in 6 of 8 (75%) of patients.		
					Ultrasound				

Noszian et	Austria	3	69	44.5 (14-75)	Ultrasound	Surgery (43) /	37 of the 43 patients who had surgery had findings in	n/a	n/a
al. (1995a)						Clinical Follow-	agreement with those on ultrasound. Of these, 6 of 39		
						Up (26)	patients were incorrectly diagnosed as having		
							displaced ligaments: 5 had intact ligaments and 1 a		
							rupture 'in-situ'. /the remaining 4 patients had non-		
							displaced ligaments correctly diagnosed.		
Schnur et	US	3	16	(17-66)	Ultrasound	Surgery (10)/	10 patients had an ultrasound diagnosis of a complete	n/a	n/a
al. (2002)						Clinical Follow-	rupture. Of these, 7 of 8 had this confirmed at surgery.		
						Up (6)	1 other patient was diagnosed with a complete rupture		
							at repeat ultrasound but found to have an attenuated		
							ligament at surgery needing repair. Another patient		
							diagnosed with an intact UCL on ultrasound was found		
							to have a giant cell tumour at surgery. 6 patients were		
							treated conservatively.		
Shekarci	Iran	1	20	38.6 (16-64)	Ultrasound	MRI	7 patients had complete UCL rupture diagnosed on	71.4%	84.6%
et al.							ultrasound of which 5 were confirmed with MRI. 13	(complete	(complete
(2018)							patients were diagnosed with an intact UCL of which	rupture)	rupture)
							11 were confirmed with MRI.		
Susic et	Denmark	1	14	Not reported	Ultrasound	Surgery	All 14 patients had ruptured ligaments at surgery.	40%	78%
al. (1999)							Ultrasound correctly diagnosed 2 of 5 completely torn	(dislocation)	(dislocation)
							and displaced ligaments and 7 of 9 undisplaced		
							ligaments. 2 patients diagnosed with displaced		

	ligaments with ultrasound were found to be	
	undisplaced.	

Table 3: Characteristics of included studies- MRI Assessment

Study	Country	Level of	Number	Mean	Index	Reference	Outcome summary	Sensitivity	Specificity
(Year)	of Origin	Evidence	of	age	Test(s)	Test(s)			
			patients	(range)					
Harper et al.	US	3	19	21.5 (21-	MRI / MR	Surgery (8) /	5 patients underwent MRI of which 2 had a diagnosis of	n/a	n/a
(1996)				46)	arthrography	Clinical Follow-	UCL dislocation confirmed at surgery. The remaining 3		
					/ stress	Up (11)	had no signs of dislocation were treated successfully		
					radiography		conservatively.		
					(SR)		14 patients underwent both MRa and SR. 6 had		
							diagnoses confirmed at surgery, 8 were treated		
							successfully conservatively.		
Hergen et	Austria	2	17	37 (14-70)	Ultrasound /	Surgery	MRI correctly diagnosed 11 undisplaced ruptures and 6	100%	100%
al. (1995b)					MRI		completely torn and displaced UCLs of which all were	(undisplaced	(undisplaced
							confirmed at surgery.	and	and
								completely	completely
								torn/displaced	torn/displaced
								ligaments)	ligaments)

Hinke et al.	US	3	11	Not	MRI	Surgery (5) /	All 11 patients were diagnosed with complete UCL	n/a	n/a
(1994)				reported		Clinical Follow-	rupture on MRI. Of these, 5 were confirmed at surgery.		
						Up (6)	6 were managed successfully with clinical follow-up. 2		
							of 3 UCL dislocations were correctly diagnosed on MRI		
							prospectively.		
Louis et al.	US	3	3	20 (19-21)	MRI	Surgery (2) /	2 patients had completely torn/displaced UCLs	n/a	n/a
(1989)						Clinical Follow-	diagnosed on MRI and confirmed at surgery. The last		
						Up (1)	had a ligamentous strain diagnosed on MRI and was		
							successfully manged conservatively.		
Milner et al.	US	3	43	39 (16-69)	MRI	Surgery (24) /	14 patients had a completely torn/displaced UCL	n/a	n/a
(2015)						Clinical Follow-	diagnosed on MRI and treated successfully with		
						Up (19)	surgery. 10 patients had a complete rupture with >3mm		
							separation on MRI. These were initially treated		
							conservatively but 9 required surgical repair. 1 of 5		
							patients with a complete rupture and <3mm separation		
							required surgery.		
Romano et	Canada	3	21	(14-62)	MRI	Surgery (10) /	In the 10 patients who underwent surgery, 7 were	n/a	n/a
al. (2003)						Clinical Follow-	correctly diagnosed by MRI as having a completely		
						Up (11)	completely torn/displaced ligament and 2 a moderately		
							displaced tear. The 1 misdiagnosed moderately		
							displaced UCL had a partial tear at surgery.		

	Of the 11 treated conservatively there were 2 partial	
	tears, 4 minimally displaced tears, 1 completely	
	torn/displaced tear, 1 moderately displaced tear and 3	
	non-displaced tears diagnosed on MRI.	

Table 4 Study characteristics

Author	Year	Journal	Setting	Type of	Populatio	Intervention	Comparator	Primary	Outcomes	Time points
				study	n			outcom		
								е		
Crowley et	2013	Technique	Single	RCT	Acute	Early active	Immobilisation in a	None	Range of motion,	Weekly first 4
al.		s in Hand &	centre,		complete	mobilisation using a	plaster-of-Paris thumb	specified	return to normal	weeks,1, 3 and 6
		Upper	Departme		UCL ruptures	custom-made	spica for 4-weeks		hand function and	months
		Extremity	nt of		repaired with	thermoplastic splint			work,	
		Surgery	Plastic		Mitek anchors				complications	
			Surgery							
Katolik et al.	2008	Plastic &	Single	Retrospective	Acute	Intraosseous suture	Pull-out suture tied over	None	Range of motion,	Endpoint only,
		Reconstru	centre,	cohort study	complete	anchor and early	a button and cast	specified	grip strength,	mean 29 (range
		ctive	Hand		UCL ruptures	mobilisation	immobilisation		pinch strength,	14 to 45) months
		Surgery	Centre		treated				patient	
					surgically				satisfaction,	
									complication	

Lane, L.B	1991	American	Single	Retrospective	Acute	Suture of UCL to	Pull-out suture with K-	None	Range of motion	Endpoint only,
		Journal of	centre	cohort study	complete	tendinous insertion of	wire fixation MCPJ -	specified	(full or partial,	mean 3.9 (range
		Sports	Orthopae		UCL ruptures	the adductor pollicis -	'Traditional' method		strength (full or	2.0 to 8.5) years
		Medicine	dic		treated	'New' method.			partial), overall	
			Surgery		surgically				outcome	
			Departme						(excellent, good	
			nt						or poor), stability,	
									pain, ability to	
									return to previous	
									level of	
									competition,	
									complications	
Rocchi et al.	2014	European	Single	RCT	Acute	Modified splint post-	Immobilisation in a	None	Pain (VAS),	1, 2, 6 and 12
		journal of	centre,		complete	operatively, allowing	hand-based	specified	Dreiser's	months
		physical &	Orthopae		UCL ruptures	flexion-extension of the	thermoplastic spica		functional hand	
		rehabilitati	dic and		treated	MCPJ	splint post-operatively,		index, range of	
		on	Hand		surgically		that immobilised the		motion, pinch	
		medicine	Surgery				MCPJ		strength, time off	
			Departme						work, residual	
			nt						symptoms,	
									incidence of	
									recurrence,	
									number of	
									physiotherapy	
									sessions,	
									complications	
Saetta et al.	1992	Journal of	Single	Retrospective	Acute	Repair using pre-	Repair using non-	None	Pain (VAS),	Endpoint only,
		Hand	centre,	cohort study	complete	fashioned pull-out steel	absorbable suture	specified	overall function	mean 19 (range 6
		Surgery -	Accident		UCL ruptures	suture			(excellent, good,	to 36) months

		British	&		treated				fair or poor),	
		Volume	Emergen		surgically				pinch, key and	
			су						grasp grip	
			Departme						strength, degrees	
			nt						of movement on	
									radial stress of the	
									MCPJ, sensation	
									in the distribution	
									of the superficial	
									radial nerve to pin-	
									prick and light	
									touch.	
Sollerman et	1991	Acta	Single	RCT	Acute UCL	Immobilisation in a	Plaster cast	None	Range of motion,	Endpoint only,
al.		Orthopaedi	centre,		ruptures	functional splint that	immobilisation.	specified	stability, pinch	mean 15 (range
		са	Hand		treated	allows flexion and			grip strength,	11 to 41) months
		Scandinavi	Surgery		surgically or	extension of the MCPJ			stability, length of	
		са	Departme		non surgically	but prevents ulnar and			sick leave	
			nt			radial deviation of the				
						thumb.				

MCPJ – Metacarpophalangeal joint. RCT – Randomised controlled trial. UCL – Ulnar collateral ligament. VAS – Visual analogue scale
Author	Year	Inclusion criteria	Exclusion criteria	Number of	Mean age	Sex of	Data comments
				participant	of	participant	
				S	participant	S	
					s		
Crowley et al.	2013	All adult patients who underwent Mitek	None specified	12	Median 42	8 males	
		bone anchor repair for a ruptured UCL.			(range 20 to	4 females	
					72) years		
Katolik et al.	2008	Complete rupture of the UCL, within 4	Avulsion fractures that	73	32 years	Not reported	
		weeks of injury.	comprised more than 10% of the				
		Diagnosis established clinically by	articular surface.				
		manual stress testing of the thumb					
		metacarpophalangeal in 30° of flexion.					
		Diagnosis was confirmed if there is					
		absolute laxity of > 30° , or laxity of 10°					
		greater than the contralateral side,					
		without evidence of a firm endpoint.					
Lane, L.B	1991	Acute grade III UCL injuries in athletes,	None specified	32	30 (range 16	20 males	Characteristics not individually
		diagnosed as laxity in excess of 35°			to 76) years	16 females	reported for the intervention
		and/or 15 more than the contralateral					groups.
		thumb with the metacarpophalangeal					
		joint in 30° of flexion.					
Rocchi et al.	2014	Acute, within 0-7 days, complete tear of	Partial suspected tear of UCL.	30	39 (range 16	24 males	Characteristics not individually
		UCL of the thumb.	Cases with associated injuries of		to 64) years	6 females	reported for the intervention
		Complete UCL tear diagnosed by clinical	the skin, tendons, nerves,				groups
		examination when there was no solid	vessels or bony fractures.				

Table 5 Details of study participants, demographics and eligibility criteria.

		endpoint at valgus stress, with more than					
		30° stressed radial deviation and more					
		than 20° difference compared to the					
		uninjured side.					
Saetta et al.	1992	Presence of a clinically unstable thumb	None specified.	25	41.4 (range 18	Not reported	Characteristics not individually
		MCPJ on radial stress following a recent			to 60) years		reported for the intervention
		injury.					groups
Sollerman et al.	1991	Acute rupture of UCL, thought to require	None specified	62	32 (range 11	43 males, 19	Some characteristics were not
		surgical treatment following clinical and			to 62) years	females	individually reported for the
		radiographic examination by experienced					intervention groups, such as age,
		orthopaedic specialists.					length of follow-up and losses to
							follow-up.

UCL – Ulnar collateral ligament. Reported results are mean (range) unless otherwise indicated.

Table 6 Details of study outcomes, time points and a summary of results.

Author	Year	Outcomes (primary in bold if	Time points	Summary of results and adverse events
		present)		
Lane, L.B	1991	Range of motion (full or partial, strength (full or	Endpoint only, mean 3.9	Patients treated using the new surgical fixation method returned to sports sooner. There
		partial), overall outcome (excellent, good or	(range 2.0 to 8.5) years	were no differences in pain, stability, range of motion and strength between the groups,
		poor), stability, pain, ability to return to previous		though ROM was restored more rapidly in the 'new' fixation group. There were 2
		level of competition, complications		complications, one re-rupture following a fall after 9 months and one incidence of broken
				pullout suture.
Crowley et al.	2013	Range of movement at MCPJ, time to return to	Weekly, then 1, 3 and 6	No significant difference in the final range of motion achieved either at the IPJ of MCPJ in
		normal function and work, complications.	months	both groups but maximum ROM was achieved earlier by the early active mobilisation group.
				Three patients in each group suffered from scar tethering, which resolved with ultrasound
				therapy and massage in all cases.
Katolik et al.	2008	Range of motion at MCPJ and IPJ, grip	Endpoint only, mean 29	Improved range of motion, reduced tourniquet time (which was used as a proxy for surgical
		strength, pinch strength, soft tissue	(range 14 to 45) months	time and thus cost) and fewer soft tissue complications were noted in the anchor group.
		complications, patient satisfaction.		There were no differences in grip strength. The authors report savings of approximately
				\$140-per-patient with the suture anchor technique.
Saetta et al.	1992	Pain (VAS), overall function (excellent, good,	Endpoint only, mean 19	Outcomes were similar across the two groups, suggesting that both methods of repair are
		fair or poor), pinch, key and grasp grip strength,	(range 6 to 36) months	equally effective.
		degrees of movement on radial stress of the		
		MCPJ, sensation in the distribution of the		
		superficial radial nerve to pin-prick and light		
		touch.		
Rocchi et al.	2014	Pain (VAS), Dreiser's functional hand index,	1, 2, 6 and 12 months	Faster and better functional results were noted in the modified splint and immediate post-
		range of motion, pinch strength, time off work,		operative immobilisation group.
		residual symptoms, incidence of recurrence,		There were three cases of temporary dysesthesia of the dorsal-ulnar region of the thumb in
		number of physiotherapy sessions,		the modified splint group and two in the standard splint group, all resolved spontaneously.
		complications.		Two cases of inflammatory scar (one in each group) were noted which resolved with anti-
				inflammatories and skin massage.

Sollerman et	1991	Range of motion, stability, pinch grip strength,	Endpoint only, mean 15	There was no difference with regards stability, range of motion, strength of the injured thumb,
al.		stability, length of sick leave	(range 11 to 41) months	and length of sick leave. However, the patients found the splint more comfortable than
				plaster cast immobilization.

MCPJ – Metacarpophalangeal joint. IPJ – Interphalangeal joint. VAS – Visual Analogue Scale. ROM – Range of motion.

Appendix 3: Quality assessments

Figure 3: Table and graph summary of methodological quality of clinical assessment trials

	Risk of Bias			Applicability Concerns				
	Patient Selection	Index Test	Reference Standard	Flow and Timing	Patient Selection	Index Test	Reference Standard	
Abrahamsson 1990	Ŧ	•	•		•	Ŧ		
Cooper 2005	?	•	•	?	•	•		
Heyman 1993	?	Ŧ	•	?	Ŧ	Ŧ	•	
Louis 1986	?	Ŧ	•		•	Ŧ	•	
Mahajan 2016	ŧ	Ŧ	?	•	•	Ŧ	•	
Murphey 1997	?	•	?	•	Ŧ	Ŧ	•	
😑 High		<mark>?</mark> U	nclea	r		+ L	w	



Table 7 - Consolidated summary of risk of bias in included studies using the Cochrane risk of bias tool for randomised controlled trials or the appropriate SIGN checklist

Study	Bias domain	Risk of bias	Overall	Justification of risk
			assessment	
			of risk of	
			bias	
Randomised	I controlled trials	•	1	
Crowley et al.,	Randomisation	Some concerns	High risk	The small sample size, inadequate randomisation method and limited description
2013	Deviations from the intended interventions			of outcome assessments limits the strength of their findings. In addition, no primary
	 effect of assignment to intervention) effect of adhering to intervention) 	Low		outcome measure was identified and there is no information regarding a prior
	Missing outcome data	Low		sample size calculation to estimate the number of participants required in order to
	Bias in measurement of the outcome	Low		reach meaningful conclusions.
	Bias in selection of the reported result	High		
		Some concerns		
Rocchi et al.,	Randomisation	Low	Some concerns	A variety of outcomes were measured, with no one specified primary outcome
2014	Deviations from the intended interventions			measure. Though the two groups were equal in size, group characteristics are not
	- effect of assignment to intervention)	Some concerns		reported and parameters such as age, gender and hand dominance are not
	 effect of adhering to intervention) Missing outcome data 	Low		reported separately each group.
	Bias in measurement of the outcome	Low		
	Bias in selection of the reported result	Low		
		Some concerns		

Sollerman et	Randomisation	Low	High	The method of randomisation is not clearly described. There was variable follow-
al., 1991	Deviations from the intended interventions			up with outcomes assessed at differing time points. No information is provided
	- effect of assignment to intervention)	High		regarding the validity or reliability of the method used for assessing comfort in
	 effect of adhering to intervention) Missing outcome data 	High		performing functional tasks. Little detail is provided regarding the standardisation
	Bias in measurement of the outcome	Low		of outcome assessments, blinding of assessors or participants, statistical analyses
	Bias in selection of the reported result	High		or a prior sample size calculations to inform number of participants required to
		Some concerns		detect statistical or clinically significant changes.
Retrospective s	tudies		I	
Katolik et al.,	Selection	Low	Acceptable	A sound methodology was applied, and the authors attempted to address the
2008	Confounding	Low	(Some	potential impact of confounding. However, it is limited by the sample size,
	Performance bias	Low	concerns)	retrospective nature and limited reporting of results, measures of uncertainty, data
	Attrition bias/missing data	High		distribution and details about between-group characteristics.
	Detection bias	Low		
	Statistical analysis	Low		
Lane, L.B,	Selection	No information	Low quality	There were differential numbers of participants per group, with only seven in the
1991	Confounding	High	(high risk of	traditional repair group. Follow-up was variable between the groups, which may
	Performance bias	No information	bias)	have an impact on outcomes, as they were assessed at differing times post
	Attrition bias/missing data	No information		intervention. There is no information provided about characteristics of the two
	Detection bias	No information		groups, and therefore it is not possible to compare the groups.
	Statistical analysis	No information		

Saetta et al.,	Selection	No information	Low	qual	ty	Inadequate information is provided about how confounding has been addressed
1992	Confounding	No information	(high	risk	of	or acknowledged. No information is provided about patient characteristics or
	Performance bias	No information	bias)			between group differences. Though outcome assessment was consistent across
	Attrition bias/missing data	High				the groups, they were performed at differing timepoints both within individuals and
	Detection bias	High				across groups.
	Statistical analysis	No information				

	R	isk o	f Bia	s	<u>Appl</u>	Applicability Concerns				
	Patient Selection	Index Test	Reference Standard	Flow and Timing	Patient Selection	Index Test	Reference Standard			
Chuter 2009		?		?	•	•	•			
Hergan 1995a	?	?	?	?	•	Ŧ	•			
Hergan 1995b	?	?	•	?	•	Ŧ	•			
Hoglund 1995	?	Ŧ	?	?	•	÷	•			
Jones 2000	?	?	?		•	Ŧ	?			
Kohut 1993	?	Ŧ	?	•	•	Ŧ	•			
Melville 2013	•	Ŧ	?	•	•	•	•			
Murphey 1997	•	•	?	•	•	•	•			
Noszlan 1995	?	Ŧ	•	•	•	Ŧ	•			
Schnur 2002	+	•	•	•	•	Ŧ	•			
Shekarchi 2018	?	Ŧ	?	?	•	Ŧ	•			
Susic 1999	+	?	Ŧ	?	•	Ŧ	•			
e High		(?) Un	clear			+ Low			

Figure 4: Table and graph summary of methodological quality of ultrasound assessment trials





Figure 5: Table and graph summary of methodological quality of MRI assessment trials



Appendix 4: Key clinical practice recommendations

- 1. Clinical examination is recommended to assess for significant laxity of the UCL (low evidence).
- 2. X-rays in orthogonal planes should be obtained to check for fractures and joint subluxation.
- 3. There is insufficient evidence to mandate the routine use of ultrasound (USS) or magnetic resonance imaging (MRI).
- 4. Patients without significant joint laxity should be treated non-surgically.
- 5. It is reasonable to offer early surgery or non-surgical immobilisation of the MCPJ to patients with significant joint laxity on clinical examination (**very low evidence**).

Appendix 5: Patient flow algorithm



Appendix 6: Support Tool: Quick reference guide

BSSH Evidence for Surgical Treatment (BEST): Evidence based management of acute ulnar collateral ligament of the thumb injuries

Key clinical practice recommendations:

- 1. Clinical examination is recommended to assess for significant laxity of the UCL (**low evidence**)
- 2. X-rays in orthogonal planes should be obtained to check for fractures and joint subluxation.
- 3. There is insufficient evidence to mandate the routine use of ultrasound (USS) or magnetic resonance imaging (MRI)
- 4. Patients without significant joint laxity should be treated non-surgically
- 5. It is reasonable to offer early surgery or non-surgical immobilisation of the MCPJ to patients with significant joint laxity on clinical examination (**very low evidence**)

Appendix 7: References

- Musharafieh RS, Bassim YR, Atiyeh BS. Ulnar collateral ligament rupture of the first metacarpophalangeal joint: a frequently missed injury in the emergency department. J Emerg Med [Internet]. 1997;15(2):193–6. Available from: http://ovidsp.ovid.com/ovidweb.cgi?T=JS&CSC=Y&NEWS=N&PAGE=fulltext&D=medc&AN=9144061
- 2. Whiting PF, Rutjes AWS, Westwood ME, Mallett S, Deeks JJ, Reitsma JB, et al. QUADAS-2: a revised tool for the quality assessment of diagnostic accuracy studies. Ann Intern Med. 2011 Oct;155(8):529–36.
- Abrahamsson SO, Sollerman C, Lundborg G, Larsson J, Egund N. Diagnosis of displaced ulnar collateral ligament of the metacarpophalangeal joint of the thumb. J Hand Surg - Am Vol [Internet]. 1990;15(3):457–60. Available from: http://ovidsp.ovid.com/ovidweb.cgi?T=JS&CSC=Y&NEWS=N&PAGE=fulltext&D=med3&AN=2348064
- 4. Cooper JG, Johnstone AJ, Hider P, Ardagh MW. Local anaesthetic infiltration increases the accuracy of assessment of ulnar collateral ligament injuries. Emerg Med Australas. 2005/03/31. 2005;17(2):132–6.
- Murphey SL, Hashimoto BE, Buckmiller J, Kramer D, Wiitala L. Ultrasonographic stress testing of ulnar collateral ligament injuries of the thumb. J Ultrasound Med [Internet]. 1997;16(3):201–7. Available from: http://ovidsp.ovid.com/ovidweb.cgi?T=JS&CSC=Y&NEWS=N&PAGE=fulltext&D=medc3&AN=9166818
- 6. Mahajan M, Tolman C, Wurth B, Rhemrev SJ. Clinical evaluation vs magnetic resonance imaging of the skier's thumb: A prospective

cohort of 30 patients. Eur J Radiol. 2016/09/27. 2016;85(10):1750-6.

- Louis DS, Huebner Jr. JJ, Hankin FM. Rupture and displacement of the ulnar collateral ligament of the metacarpophalangeal joint of the thumb. Preoperative diagnosis. J Bone Jt Surg Am. 1986/12/01. 1986;68(9):1320–6.
- Heyman P, Gelberman RH, Duncan K, Hipp JA. Injuries of the ulnar collateral ligament of the thumb metacarpophalangeal joint.
 Biomechanical and prospective clinical studies on the usefulness of valgus stress testing. Clin Orthop Relat Res [Internet]. 1993;(292):165– 71. Available from: http://ovidsp.ovid.com/ovidweb.cgi?T=JS&CSC=Y&NEWS=N&PAGE=fulltext&D=med3&AN=8519106
- 9. Louis DS, Huebner Jr. JJ, Hankin FM. Rupture and displacement of the ulnar collateral ligament of the metacarpophalangeal joint of the thumb. Preoperative diagnosis. J Bone Jt Surg Am. 1986/12/01. 1986;68(9):1320–6.
- 10. Shekarchi B, Mokhdanzadeh Dashti M, Shahrezaei M, Karimi E. The Accuracy of Ultrasonography in Detection of Ulnar Collateral Ligament of Thumb Injuries; a Cross-Sectional Study. Emerg (Tehran, Iran) [Internet]. 2018;6(1):e15. Available from: http://ovidsp.ovid.com/ovidweb.cgi?T=JS&CSC=Y&NEWS=N&PAGE=fulltext&D=prem&AN=29503840
- Susic D, Hansen BR, Hansen TB. Ultrasonography may be misleading in the diagnosis of ruptured and dislocated ulnar collateral ligaments 11. Plast Scand Reconstr Surg Hand Surg [Internet]. 1999;33(3):319-20. of the thumb. J Available from: http://ovidsp.ovid.com/ovidweb.cgi?T=JS&CSC=Y&NEWS=N&PAGE=fulltext&D=medc&AN=10505446
- 12. Chuter GS, Muwanga CL, Irwin LR. Ulnar collateral ligament injuries of the thumb: 10 years of surgical experience. Injury [Internet]. 2009;40(6):652–6. Available from:

http://ovidsp.ovid.com/ovidweb.cgi?T=JS&CSC=Y&NEWS=N&PAGE=fulltext&D=med6&AN=19389670

- Hergan K, Mittler C, Oser W. Ulnar collateral ligament: differentiation of displaced and nondisplaced tears with US and MR imaging. Radiology. 1995/01/01. 1995;194(1):65–71.
- 14. Melville D, Jacobson JA, Haase S, Brandon C, Brigido MK, Fessell D. Ultrasound of displaced ulnar collateral ligament tears of the thumb: the Stener lesion revisited. Skeletal Radiol [Internet]. 2013;42(5):667–73. Available from: http://ovidsp.ovid.com/ovidweb.cgi?T=JS&CSC=Y&NEWS=N&PAGE=fulltext&D=med9&AN=23001117
- 15. Hoglund M, Tordai P, Muren C. Diagnosis by ultrasound of dislocated ulnar collateral ligament of the thumb. Acta radiol [Internet]. 1995;36(6):620–5. Available from: http://ovidsp.ovid.com/ovidweb.cgi?T=JS&CSC=Y&NEWS=N&PAGE=fulltext&D=med3&AN=8519573
- 16. Kohut M, Droz CP, Della Santa DR. Palmar dislocation of the metacarpophalangeal joint of the thumb. About one case treated straightaway by bilateral suture of the ligaments, reinforced by a tendinoplasty. [French]. Ann Chir la Main du Memb Super [Internet]. 1994;13(1):50–5. Available from: http://ovidsp.ovid.com/ovidweb.cgi?T=JS&CSC=Y&NEWS=N&PAGE=fulltext&D=emed5&AN=24082880
- 17. Jones MH, England SJ, Muwanga CL, Hildreth T. The use of ultrasound in the diagnosis of injuries of the ulnar collateral ligament of the thumb. J Hand Surg Br. 2000/04/14. 2000;25(1):29–32.
- Noszian IM, Dinkhauser LM, Straub GM, Orthner E. Ulnar collateral ligament injuries of the thumb. Dislocation caused by stress radiography in 2 cases. Acta Orthop Scand [Internet]. 1995;66(2):156–7. Available from: http://ovidsp.ovid.com/ovidweb.cgi?T=JS&CSC=Y&NEWS=N&PAGE=fulltext&D=medc3&AN=7740947

- 19. Schnur DP, DeLone FX, McClellan MR, Bonavita J, Witham RS. Ultrasound: A powerful tool in the diagnosis of ulnar collateral ligament injuries of the thumb. Ann Plast Surg [Internet]. 2002;49(1):19–23. Available from: http://ovidsp.ovid.com/ovidweb.cgi?T=JS&CSC=Y&NEWS=N&PAGE=fulltext&D=emed7&AN=34755462
- Harper MT, Chandnani VP, Spaeth J, Santangelo JR, Providence BC, Bagg MA. Gamekeeper thumb: diagnosis of ulnar collateral ligament injury using magnetic resonance imaging, magnetic resonance arthrography and stress radiography. J Magn Reson Imaging [Internet].
 1996;6(2):322–8. Available from: http://ovidsp.ovid.com/ovidweb.cgi?T=JS&CSC=Y&NEWS=N&PAGE=fulltext&D=medc&AN=9132097
- 21. Hinke DH, Erickson SJ, Chamoy L, Timins ME. Ulnar collateral ligament of the thumb: MR findings in cadavers, volunteers, and patients with ligamentous injury (Gamekeeper's thumb). Am J Roentgenol [Internet]. 1994;163(6):1431–4. Available from: http://ovidsp.ovid.com/ovidweb.cgi?T=JS&CSC=Y&NEWS=N&PAGE=fulltext&D=emed6&AN=25002332
- 22. Louis DS, Buckwalter KA. Magnetic resonance imaging of the collateral ligaments of the thumb. J Hand Surg Am Vol [Internet].1989;14(4):739–41.Availablefrom:

http://ovidsp.ovid.com/ovidweb.cgi?T=JS&CSC=Y&NEWS=N&PAGE=fulltext&D=medc3&AN=2754211

- 23. Milner CS, Manon-Matos Y, Thirkannad SM. Gamekeeper's thumb--a treatment-oriented magnetic resonance imaging classification. J Hand Surg Am. 2014/10/11. 2015;40(1):90–5.
- 24. Romano WM, Garvin G, Bhayana D, Chaudhary O. The spectrum of ulnar collateral ligament injuries as viewed on magnetic resonance imaging of the metacarpophalangeal joint of the thumb. Can Assoc Radiol J [Internet]. 2003;54(4):243–8. Available from:

http://ovidsp.ovid.com/ovidweb.cgi?T=JS&CSC=Y&NEWS=N&PAGE=fulltext&D=medc3&AN=14593774

- 25. Sollerman C, Abrahamsson SO, Lundborg G, Adalbert K. Functional splinting versus plaster cast for ruptures of the ulnar collateral ligament of the thumb. A prospective randomized study of 63 cases. Acta Orthop Scand [Internet]. 1991;62(6):524–6. Available from: http://ovidsp.ovid.com/ovidweb.cgi?T=JS&CSC=Y&NEWS=N&PAGE=fulltext&D=med3&AN=1767639
- 26. Rocchi L, Merolli A, Morini A, Monteleone G, Foti C. A modified spica-splint in postoperative early-motion management of skier's thumb lesion: a randomized clinical trial. Eur J Phys Rehabil Med [Internet]. 2014;50(1):49–57. Available from: http://ovidsp.ovid.com/ovidweb.cgi?T=JS&CSC=Y&NEWS=N&PAGE=fulltext&D=med8&AN=24185690
- 27. Crowley TP, Stevenson S, Taghizadeh R, Addison P, Milner RH. Early active mobilization following UCL repair With Mitek bone anchor.
 Tech Hand Up Extrem Surg [Internet]. 2013;17(3):124–7. Available from: http://ovidsp.ovid.com/ovidweb.cgi?T=JS&CSC=Y&NEWS=N&PAGE=fulltext&D=med7&AN=23970193
- 28. Saetta JP, Phair IC, Quinton DN. Ulnar collateral ligament repair of the metacarpo-phalangeal joint of the thumb: a study comparing two methods of repair. J Hand Surg Br Vol [Internet]. 1992;17(2):160–3. Available from: http://ovidsp.ovid.com/ovidweb.cgi?T=JS&CSC=Y&NEWS=N&PAGE=fulltext&D=med3&AN=1534110
- 29. Lane LB. Acute Grade III ulnar collateral ligament ruptures: a new surgical and rehabilitation protocol. Am J Sports Med [Internet]. 1991;19(3):234–8. Available from: http://articles.sirc.ca/search.cfm?id=276721
- 30. Katolik LI, Friedrich J, Trumble TE. Repair of acute ulnar collateral ligament injuries of the thumb metacarpophalangeal joint: a retrospective

comparison of pull-out sutures and bone anchor techniques. Plast Reconstr Surg [Internet]. 2008;122(5):1451–6. Available from: http://ovidsp.ovid.com/ovidweb.cgi?T=JS&CSC=Y&NEWS=N&PAGE=fulltext&D=med6&AN=18971729

- 31. Pichora DR, McMurtry RY, Bell MJ. Gamekeepers thumb: A prospective study of functional bracing. J Hand Surg Am [Internet]. 1989;14(3):567–73. Available from: http://www.sciencedirect.com/science/article/pii/S0363502389800267
- Landsman JC, Seitz Jr. WH, Froimson AI, Leb RB, Bachner EJ. Splint immobilization of gamekeeper's thumb. Orthopedics. 1995/12/01.
 1995;18(12):1161–5.